# UC Davis UC Davis Previously Published Works

#### Title

Vitamin A supplementation boosts control of antibiotic-resistant Salmonella infection in malnourished mice

**Permalink** https://escholarship.org/uc/item/9f13m01g

**Journal** PLOS Neglected Tropical Diseases, 14(10)

**ISSN** 1935-2727

#### **Authors**

Stull-Lane, Annica R Lokken-Toyli, Kristen L Diaz-Ochoa, Vladimir E <u>et al.</u>

#### **Publication Date**

2020

### DOI

10.1371/journal.pntd.0008737

Peer reviewed

Check for updates

# 

**Citation:** Stull-Lane AR, Lokken-Toyli KL, Diaz-Ochoa VE, Walker GT, Cevallos SA, Winter ALN, et al. (2020) Vitamin A supplementation boosts control of antibiotic-resistant *Salmonella* infection in malnourished mice. PLoS Negl Trop Dis 14(10): e0008737. https://doi.org/10.1371/journal. pntd.0008737

**Editor:** Alfredo G. Torres, University of Texas Medical Branch, UNITED STATES

Received: April 9, 2020

Accepted: August 20, 2020

Published: October 2, 2020

**Copyright:** © 2020 Stull-Lane et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript and its Supporting Information files.

**Funding:** This project was supported by NIH/NIAID grants AI098078 & AI128886. ARSL was supported by NIH/NIAID grant F30AI143146, and the National Center for Advancing Translational Sciences, National Institutes of Health, through grant number UL1 TR001860 and linked award TL1 TR001861. KLLT was supported by the

RESEARCH ARTICLE

# Vitamin A supplementation boosts control of antibiotic-resistant *Salmonella* infection in malnourished mice

Annica R. Stull-Lane<sup>1</sup>, Kristen L. Lokken-Toyli<sup>1,2</sup>, Vladimir E. Diaz-Ochoa<sup>1</sup>, Gregory T. Walker<sup>1</sup>, Stephanie A. Cevallos<sup>1</sup>, Andromeda L. N. Winter<sup>1</sup>, Ariel Del Hoyo Muñoz<sup>1</sup>, Guiyan G. Yang<sup>1,3</sup>, Eric M. Velazquez<sup>1</sup>, Chun-Yi Wu<sup>4</sup>, Renée M. Tsolis<sup>1</sup>\*

1 Department of Microbiology & Immunology, School of Medicine, University of California Davis, Davis, California, United States of America, 2 Department of Microbiology, New York University, New York, New York, United States of America, 3 Department of Veterinary Clinical Sciences, College of Veterinary Medicine, China Agricultural University, Beijing, China, 4 Department of Neurology, School of Medicine, University of California, Davis, Sacramento, California, United States of America

\* rmtsolis@ucdavis.edu

## Abstract

Disseminated disease from non-typhoidal Salmonella enterica strains results in >20% mortality globally. Barriers to effective treatment include emerging multidrug resistance, antibiotic treatment failure, and risk factors such as malnutrition and related micronutrient deficiencies. Individuals in sub-Saharan Africa are disproportionately affected by nontyphoidal S. enterica bloodstream infections. To inform a clinical trial in people, we investigated vitamin A as a treatment in the context of antibiotic treatment failure in a mouse model of vitamin A deficiency. Vitamin A-deficient (VAD) mice exhibited higher systemic bacterial levels with a multidrug-resistant clinical isolate in comparison to mice on a control diet. Sexspecific differences in vitamin A deficiency and disseminated infection with S. enterica serotype Typhimurium (S. Typhimurium) were observed. VAD male mice had decreased weight gain compared to control male mice. Further, infected VAD male mice had significant weight loss and decreased survival during the course of infection. These differences were not apparent in female mice. In a model of disseminated S. Typhimurium infection and antibiotic treatment failure, we assessed the potential of two consecutive doses of vitamin A in alleviating infection in male and female mice on a VAD or control diet. We found that subtherapeutic antibiotic treatment synergized with vitamin A treatment in infected VAD male mice, significantly decreasing systemic bacterial levels, mitigating weight loss and improving survival. These results suggest that assessing vitamin A as a therapy during bacteremia in malnourished patients may lead to improved health outcomes in a subset of patients, especially in the context of antibiotic treatment failure.

#### Author summary

Non-typhoidal *Salmonella* serotypes generally cause diarrhea in people. However, there are certain factors that make people at risk of developing a more severe infection where

American Heart Association Predoctoral Fellowship 15PRE21420011 and NIH/NIAID grant F32AI143043. GTW, SAC, & ARSL were supported by the Animal Models of Infectious Diseases Training Program NIH/NIAID grant T32AI060555. ALNW was supported by the McNair Scholars Program at UC Davis. ADHM was supported by the UC Davis Maximizing Access to Research Careers (MARC) Program (NIGMS-MARC-U-STAR grant GM083894). GGY was supported by China Scholarship Council. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

the bacteria can enter the blood and cause fever and whole-body symptoms. Patients with this infection are usually hospitalized, and about one in five patients do not survive. The factors that make this bloodstream infection possible include pathogen features like resistance to antibiotics and patient factors like a malnourished state. Better treatments are needed. In this study, the authors assess vitamin A as a treatment during antibiotic treatment failure in a mouse model. Vitamin A-deficient male mice have better outcomes with vitamin A and antibiotic co-therapy, whereas female mice do not benefit. Despite similar levels of bacteria causing infection systemically, female mice show better outcomes in terms of weight loss and survival than male mice overall. This research provides evidence that a clinical study assessing vitamin A as a treatment in people could lead to improved survival for malnourished patients presenting with severe bloodstream infection.

#### Introduction

In susceptible populations globally, there is a vicious cycle of infection and malnutrition [1]. Malnutrition overlaps with deficiencies in micronutrients, such as iron, iodine, zinc and fatsoluble vitamins like vitamin A [2]. Severe or recurrent infections can lead to vitamin A deficiency due to impaired nutrient absorption and decreased food intake, as well as direct nutrient loss through urine [3, 4]. In turn, vitamin A deficiency compromises the gastrointestinal mucosal epithelial barrier, increasing susceptibility to enteroinvasive bacteria and sepsis [4]. One of the most common enteric pathogens is Salmonella enterica [5]. Non-typhoidal serotypes of S. enterica cause significant morbidity and mortality worldwide, resulting in 93.8 million cases of gastroenteritis and 155,000 associated deaths annually [6]. Invasive non-typhoidal Salmonella (iNTS) infection, a severe febrile illness with bacteremia, disproportionately affects infants, older adults and immunosuppressed individuals. An estimated 3.4 million cases of invasive disease occur globally per year, resulting in 680,000 deaths [7]. Malnutrition and concurrent malaria are key risk factors for systemic disease, and populations in sub-Saharan Africa are disproportionately affected [8]. Barriers to effective treatment of iNTS illness include appropriate diagnostic tools, antibiotic treatment failure and antimicrobial resistance [9]. Notably, the mortality rate due to iNTS in children reaches 20–25% [10], indicating that more effective treatment is vital.

Vitamin A administration has been studied as treatment of infectious disease, such as for measles in children [11]. In addition, the WHO recommends vitamin A supplementation in the treatment of children that present with severe acute malnutrition [12]. Given the importance of vitamin A in immune function [4, 13, 14], these treatment recommendations suggest a role for vitamin A as a host-directed therapy [15]. However, vitamin A has not been studied as a treatment for nonspecific febrile illnesses like iNTS disease, especially in the context of antibiotic treatment failure.

This study aimed to explore the therapeutic potential of vitamin A administered after onset of infection, simulating care of a sick patient presenting to clinic. Our hypothesis was that vitamin A therapy could boost the effect of subtherapeutic antibiotic treatment and improve treatment outcomes for a drug-resistant iNTS infection. To address this question, we used mice to model antibiotic treatment failure of *S. enterica* serotype Typhimurium in the setting of vitamin A deficiency.

#### Materials and methods

#### **Ethics statement**

Mouse experiments were carried out in compliance with the National Institutes of Health policies on animal welfare, the Animal Welfare Act, and all other applicable federal, state and local laws. Mice were cared for by research staff and University of California (UC) Davis Teaching and Research Animal Care Services (TRACS) under a program accredited by the Association for Assessment and Accreditation of Laboratory Animal Care International (AAALAC). TRACS maintains a health monitoring program administered through the UC Davis Comparative Pathology Laboratory. When needed, Campus Veterinary Services is available to provide interventional and supportive care. All mouse experiments were approved by the UC Davis Institutional Animal Care and Use Committee (IACUC) under protocol 19900.

#### Animals

Inbred C57BL/6J-*Slc11a1*<sup>+/+</sup> mice are a C57BL/6J congenic strain containing a genomic segment with *Slc11a1* from S29S1 mice that was introgressed into Chromosome 1 [16]. This mouse line was maintained at UC Davis under specific pathogen-free conditions in cages with sterile ALPHA-dri bedding manufactured by Shepherd Specialty Papers (Watertown, TN). To generate VAD and control mice, C57BL/6J-*Slc11a1*<sup>+/+</sup> dams were given a VAD diet at 2 weeks of gestation. Pups were placed on either VAD or control diets at weaning and were maintained as groups throughout each experiment. Mouse weights were monitored weekly during growth. Custom VAD and control diets were obtained from Envigo Teklad Diets (Madison, WI). When the model was established, liver retinol concentration was confirmed by HPLC. Liver retinol concentration of control mice ranged from 120–300 nmol/g, whereas liver retinol concentration of VAD mice ranged from 0–7 nmol/g. Inbred C57BL/6J mice were purchased from The Jackson Laboratory (Bar Harbor, ME) and were maintained on standard chow.

#### In vitro minimum inhibitory concentration experiment

D23580, a multidrug-resistant *Salmonella enterica* serotype Typhimurium clinical bloodstream isolate, was received from Robert Heyderman at the Malawi-Liverpool-Wellcome Trust Clinical Research Centre [17]. Inocula were cultured aerobically in Luria-Bertani (LB) media with 30 µg/ml chloramphenicol shaking for 16–18 hours at 37°C, subcultured to a calculated optical density (OD) of 0.001, and grown 16–18 hours rocking at 37°C in a 96-well plate containing LB with serial dilutions of enrofloxacin ranging from 0 µg/ml to 2.56 µg/ml. Enrofloxacin (Baytril 100 by Bayer) was obtained from the Pharmaceutical Service at the UC Davis Veterinary Medical Teaching Hospital. The OD was read at 595 nm on a BioRad Model 680 Microplate Reader (Hercules, CA). The experiment was repeated six times, and the average minimum inhibitory concentration (MIC) was recorded. A determination of susceptible (S), intermediate (I) or resistant (R) was made according to CLSI guidelines [18].

#### In vivo infection with Salmonella enterica serotype Typhimurium

Inocula of S. Typhimurium strains D23580, a multilocus sequence type (ST) 313 strain isolated from blood, SL1344, an ST19 isolate from a calf with salmonellosis [19] and SARA16, an ST19 human isolate and reference strain [20, 21], were cultured in LB with shaking for 16–18 hours at 37°C. The identity of each strain was confirmed by antibiotic resistance profiling. Mice received 100  $\mu$ l of sterile PBS or 100  $\mu$ l of 1000 colony-forming units (CFU) diluted in sterile PBS by intraperitoneal (i.p.) injection. Mouse weights were monitored daily during infection. Mice were euthanized if they showed signs of lethal morbidity. Systemic bacterial levels were

characterized at day 1, 4 or 5 post-infection by determining tissue loads of S. Typhimurium (CFU). Liver and spleen were collected in PBS, weighed and homogenized using an Ultra Turrax T25 Basic mixer from IKA (Staufen, Germany). Blood was collected in K2 EDTA microtainer tubes (Becton, Dickinson and Company, Franklin Lakes, NJ), kept on ice for one hour, and then incubated for 10 minutes with 100–200  $\mu$ l of 1% Triton X-100. Liver, spleen and blood samples were serially diluted and plated on LB agar plates containing appropriate antibiotic selection. CFU/gram tissue or CFU/ml was calculated after overnight growth at 37°C.

#### In vivo treatment experiments

Mice were infected with S. Typhimurium D23580 as previously described. For preliminary antibiotic treatment experiments, S. Typhimurium systemic bacterial levels (CFU) were assessed for male (n = 5) and female (n = 5) mice on a standard diet 5 days post-infection for the following treatment groups: 0 mg/ml, 0.01 mg/ml, 0.05 mg/ml, and 0.10 mg/ml enrofloxacin delivered in the drinking water. Water and mouse weights were collected daily, and the average antibiotic consumed (µg/kg/day) was determined. For co-treatment experiments with mice on special diets, 4-6 mice were assessed for control and VAD male and female mice in the following groups: mock-treated, mock-treated and enrofloxacin, vitamin A only, and vitamin A and enrofloxacin co-treatment. Mock-treated mice received 100 µl of sterile PBS administered by oral gavage on day 1 and day 2 post-infection. Vitamin A-treated mice received 600 IU of retinyl palmitate (Interplexus, Inc., Kent, WA) in 100 µl of sterile PBS administered by oral gavage on day 1 and day 2 post-infection and were changed to control diet at day 1 post-infection. Enrofloxacin (0.05 mg/ml or 0.01 mg/ml) was administered at day 2 post-infection in the drinking water. Enrofloxacin (Baytril 100 by Bayer) was obtained from the Pharmaceutical Service at the UC Davis Veterinary Medical Teaching Hospital. Water weights were monitored daily to determine average antibiotic consumption per mouse. CFU/ gram tissue or CFU/ml was calculated for liver, spleen and blood. For measurement of plasma enrofloxacin concentrations, blood was first centrifuged for 10 minutes at 1000 x g. Plasma was stored at -80°C for further analysis by liquid chromatography with tandem mass spectrometry (LC-MS/MS). The remaining blood sample was incubated at room temperature for 10 minutes with 100–200 µl of 1% Triton X-100 and plated for CFU as previously described.

#### Plasma antibiotic measurements

LC-MS/MS was used to quantify enrofloxacin and ciprofloxacin concentrations in the mouse plasma. First, a mixed master stock solution of 10 µg/ml of enrofloxacin (MilliporeSigma, St. Louis, MO) and ciprofloxacin (MilliporeSigma, St. Louis, MO) was made in control CD-1 mouse plasma (Bioreclamation IVT, Westbury, NY) with K2 EDTA. This stock solution was then further diluted with the same control mouse plasma to establish a mouse plasma calibration curve. The following calibrator concentrations were used: 0, 1, 2.5, 5, 10, 50, 100, 500, 1000 and 5000 ng/ml. Calibrators 10, 100 and 1000 ng/ml also served as quality control (QC) samples. Moxifloxacin (MilliporeSigma, St. Louis, MO) in acetonitrile (ACN) (Fisher Scientific, Hampton, NH) was used as an internal standard such that 160  $\mu$ l of 200 ng/ml moxifloxacin in ACN was added to 40 µl of the mouse plasma calibrators, QC samples and the experimental mouse plasma samples collected as previously described. All samples were vortexed for 10 seconds. Samples were then centrifuged at room temperature for 5 minutes at 17,000 x g, and 25 µl of resulting supernatant was diluted with 175 µl of water. Next, 2 µl of the final solution was injected into the ACQUITY Ultra Performance Liquid Chromatography (UPLC) System with a BEH C18 column, 1.7 µm, 2.1 mm x 50 mm (Waters Corporation, Milford, MA). The following mobile phases were used: A ( $H_2O$  with 0.1% formic acid), and B

(ACN with 0.1% formic acid) (Fisher Scientific, Hampton, NH). Mobile phase A was also used for purging, and ACN/H<sub>2</sub>O at 50/50 (v/v) was used for the needle washing between each injection. The flow rate was 0.2 ml/min, the column temperature was set at 50°C and the autosampler temperature was set at 10°C. The following UPLC gradient program was used for the separation: 0-1 min, 10% B; 2.5 min, 72.5% B; 2.51-4.75 min, 95% B; 4.76-7 min, 10% B. The output of the UPLC was fed to a Xevo TQ-S Triple Quadrupole Mass Spectrometry (MS/MS) system (Waters Corporation, Milford, MA), which was used to ionize target molecules with electrospray ionization (ESI+) and monitor the ion m/z fragmentation transitions from 360.1-245.2 for enrofloxacin quantification, 332.2-245.2 for ciprofloxacin quantification and 402.2-261.1 for moxifloxacin quantification in multiple reaction monitoring (MRM) mode. The retention times were 2.45, 2.38 and 2.58 minutes for enrofloxacin, ciprofloxacin and moxifloxacin, respectively. The calibration curve was fitted with a weighted  $(1/x^2)$  least-squares linear regression algorithm. The detection range was from 1 ng/ml to 5000 ng/ml for enrofloxacin and from 2.5 ng/ml to 5000 ng/ml for ciprofloxacin. The extraction yield was about 80-100% for all three antibiotics and the matrix effect enhanced both enrofloxacin and ciprofloxacin MS signals by 10% and 4%, respectively. The matrix effect enhanced moxifloxacin MS signals by 48%. Both inter- and intra-batch accuracy were lower than 10% (% deviation) except that ciprofloxacin had 11% of inter-batch deviation at 2.5 ng/ml. Both intra- and inter-batch precision were also lower than 10% coefficient of variation (CV), except that enrofloxacin and ciprofloxacin had 20-40% CV at 1 ng/ml and 2.5 ng/ml, respectively. The Lower Limit of Quantification (LLOQ) for enrofloxacin was determined to be 1 ng/ml and the LLOQ for ciprofloxacin was determined to be 2.5 ng/ml.

#### Survival curves

To assess survival in untreated mice, control and VAD male and female mice were infected with S. Typhimurium D23580 as previously described, and weights were monitored daily for up to 4 days post-infection. Percent survival for each group (n = 8-9) was reported with a Kaplan-Meier survival curve. To assess male-specific survival with treatment conditions, control and VAD male mice were infected with S. Typhimurium as previously described. Control mice and one group of VAD mice were mock-treated with PBS. Other VAD groups received either enrofloxacin treatment alone (0.01 mg/ml) or co-treatment of enrofloxacin and vitamin A as previously described. Percent survival of 6–16 mice/group at 8 days post-infection was assessed with a Kaplan-Meier survival curve. To assess treatment of vitamin A alone, control and VAD male and female mice were infected with S. Typhimurium as previously described. Control mice were mock-treated with PBS, one group of VAD mice was treated with vitamin A as previously described and a second group of VAD mice received mock-treatment. Percent survival of 12 mice/group at 13 days post-infection was assessed with a Kaplan-Meier survival curve. For all survival curve experiments, mice were weighed daily and humanely euthanized when they approached 20% weight loss or displayed signs of lethal morbidity, according to our institution's humane endpoints policy.

#### Statistical analysis

Statistical analyses were performed with GraphPad Prism 8 (GraphPad, La Jolla, CA). Since data were found not to have a normal distribution, nonparametric analyses were utilized. Significance of differences between two groups was determined with a Mann-Whitney test. Significance of differences between multiple groups from a control group was determined with a Kruskal-Wallis test and a *post-hoc* Dunn's multiple comparisons test. For the survival curves, statistical significance was determined using a log-rank (Mantel-Cox) test. Data for weights,

systemic bacterial levels, antibiotic consumed and plasma enrofloxacin concentration were reported as mean  $\pm$  SEM. A p<0.05 was considered significant.

#### Results

#### Systemic Salmonella levels are higher in VAD mice

In order to model how malnutrition and associated vitamin A deficiency are risk factors for developing systemic *S*. Typhimurium infection in people, we established a mouse model of vitamin A deficiency and disseminated disease (Fig 1A). Mice were maintained on either a VAD diet or a control diet containing adequate vitamin A. To model systemic infection, mice were infected with *S*. Typhimurium D23580 via the intraperitoneal (i.p.) route. At 4d post-infection, systemic levels of *S*. Typhimurium in liver, spleen and blood were significantly higher in VAD mice compared to control (Fig 1B), and there was no sex difference observed in bacterial colonization (S1A Fig). In a similar experiment with a necropsy at 5d post-infection, no sex differences in systemic bacterial burden was observed in control mice (S1B Fig). When combining data from mice necropsied at 4d and 5d post-infection, VAD mice of both sexes still had significantly higher systemic bacterial levels than control mice (S1C Fig). These



**Fig 1. Systemic burden of** *Salmonella* **is higher in vitamin A-deficient mice.** To generate vitamin A-deficient (VAD) mice, C57BL/6 *Slc11a1*<sup>+/+</sup> pregnant mice were put on a VAD diet 2 weeks into gestation (**A**). At 3 weeks, pups were weaned onto either a VAD or control diet. Male (n = 7) and female (n = 7-8) mice on each diet were infected with *S*. Typhimurium (*S*Tm) D23580 via the intraperitoneal (i.p.) route at 9 weeks of age and systemic bacterial burden was characterized by colony-forming units (CFU) in liver, spleen and blood collected at necropsy (Nx) 4 days after infection (**B**). Data represent mean ± SEM. Significance between control and VAD mice was determined with a Mann-Whitney test of log-transformed values. A p<0.05 was considered significant.

https://doi.org/10.1371/journal.pntd.0008737.g001

data demonstrated that our model of vitamin A deficiency could be utilized to study its effect on susceptibility to systemic *S*. Typhimurium infection.

# Increased effect of VAD on S. Typhimurium-induced morbidity in male mice

It has been known for decades that vitamin A deficiency affects males and females differently [22]. To assess if this held true for our mouse model, both male and female mice were included. During maintenance on VAD and control diets, mice were weighed weekly. While diet had no effect on weight gain of female mice, VAD male mice gained significantly less weight than control mice starting at postnatal week 5 (Fig 2A). During the course of infection with *S*. Typhimurium D23580, VAD male mice had significant weight loss from day 0 to day 4 of infection, and 50% of mice in the VAD male group exhibited lethal morbidity by day 4 (Fig 2B and 2C). In contrast, VAD females and control mice of both sexes maintained their weight over the course of the experiment (Fig 2C).

While vitamin A deficiency had sex-specific effects on infection-induced morbidity (Fig 2B and 2C), both male and female mice were similar in exhibiting significantly higher systemic burdens of *S*. Typhimurium in VAD compared to control mice (S1C Fig). These results indicated that while vitamin A deficiency compromised control of systemic *S*. Typhimurium infection in both sexes, male mice were affected more strongly, both by reduced growth while on the VAD diet and by increased morbidity during infection.

#### Sex-specific effects of vitamin A deficiency on responses to S. Typhimurium are apparent early after infection

Since there was no sex difference in systemic bacterial burden by day 4 post-infection (Fig 1B) but VAD males lost weight and had decreased survival compared to female mice (Fig 2), we investigated whether an earlier time point would yield sex differences in *Salmonella* pathogenesis. Further, we assessed if this trend would hold true for multiple strains of *S*. Typhimurium, including D23580 (an ST313 strain), and the ST19 strains SL1344 and SARA16. There was no sex difference in systemic bacterial burden for any of the *S*. Typhimurium strains in mice fed a control diet (Fig 3A). However, at day 1 post-infection, male mice on a VAD diet had significantly more or a trend towards a higher systemic burden of all three *S*. Typhimurium strains than VAD female mice (Fig 3B). These results suggest that vitamin A deficiency has a more marked effect in male mice on responses that control systemic replication of *S*. Typhimurium in the early stages of infection.

#### Antibiotic treatment failure occurs at specific concentrations in vivo

Emerging multidrug-resistant *S*. Typhimurium strains like D23580 are resistant to the firstline drugs for iNTS: ampicillin, chloramphenicol and trimethoprim/sulfamethoxazole. For this reason, the fluoroquinolone ciprofloxacin is one of the currently recommended treatments for iNTS [23]. However, emergence of fluoroquinolone resistance in *Salmonella* strains and host factors like malnutrition can contribute to treatment failure [24, 25]. We chose to model antibiotic treatment failure with the fluoroquinolone enrofloxacin since it can be administered orally through the drinking water to mice and because we could measure it in the bloodstream. We first confirmed susceptibility of D23580 to enrofloxacin *in vitro*. The average MIC was 0.04 µg/ml, which by CLSI guidelines is susceptible [18]. We next sought to determine an *in vivo* MIC. In order to assess the subtherapeutic concentration of enrofloxacin administered in the drinking water, systemic bacterial levels were assessed for male and female mice at a range



**Fig 2. Increased weight loss and decreased survival seen in VAD male mice infected with** *Salmonella*. Weight (g) of VAD and control male and female mice was recorded weekly (A) during growth from 4 weeks to 9 weeks (n = 26-30). Nine week-old mice were infected with *S.* Typhimurium D23580 via the intraperitoneal (i.p.) route and weight (g) was monitored daily (B) for up to 4 days post-infection (n = 8-9). Data are represented as mean ± SEM. Significance between weights by diet or day was determined using a Mann-Whitney test (\*, p<0.05; ns, not significant). (C) Percent survival was assessed for all groups over the 4-day infection with a Kaplan-Meier survival curve (n = 8-9). Results are reported as percent survival. Pairwise comparisons between the VAD male group and each other group was determined with a survival analysis log-rank (Mantel-Cox) test (\*, p<0.05). All data represent a compilation of data from three independent experiments.

of concentrations at day 5 post-infection. To model antibiotic treatment failure, enrofloxacin administration was delayed to day 2 post-infection. We used C57BL/6J mice fed a standard diet for titration experiments. No sex difference was observed in systemic bacterial levels (S2 Fig). A significant decrease in bacteria in liver, spleen and blood was seen at an enrofloxacin concentration of 0.10 mg/ml (Fig 4A). The concentration 0.01 mg/ml was subtherapeutic, as it failed to significantly decrease systemic *S*. Typhimurium levels. The concentration 0.05 mg/ml was partly subtherapeutic, as it did not significantly decrease *S*. Typhimurium levels in the liver. Water weights and mouse weights were monitored daily during infection, and the



**Fig 3.** *Salmonella* strains colonize systemically at higher levels in VAD male mice early in infection. Systemic bacterial burden was characterized by CFU in liver, spleen and blood collected at necropsy 1 day after infection with *S*. Typhimurium D23580, SL1344 or SARA16 via the i.p. route for male and female mice (n = 4-8) on either control (A) or VAD (B) diet. Data represent mean ± SEM. Significance between male and female mice was determined with a Mann-Whitney test of log-transformed values (\*, p < 0.05; ns, not significant).

average amount of enrofloxacin consumed was calculated in  $\mu$ g/kg/day for days 2–3, 3–4 and 4–5. Enrofloxacin consumption tracked with dosage, although 0.01 mg/ml and 0.05 mg/ml were not statistically significant from the untreated group (Fig 4B). Taken together, these data demonstrated that 0.01 mg/ml and 0.05 mg/ml could model antibiotic treatment failure and thus were used for subsequent experiments with mice on special diets.

# Vitamin A and subtherapeutic antibiotic co-treatment synergize to decrease systemic S. Typhimurium levels and improve survival in VAD male mice

To assess the therapeutic potential of vitamin A in the context of antibiotic treatment failure, systemic levels of *S*. Typhimurium D23580 at day 4 post-infection were assessed for male and female mice on either control or VAD diets with the following treatment groups: mock-treated, vitamin A only, enrofloxacin (0.05 mg/ml) only, and vitamin A and enrofloxacin co-treatment (Fig 5A). In VAD male mice, co-treatment significantly decreased bacterial levels at all three systemic sites in comparison to mock treatment; however, individual treatments did not (Fig 5B). Systemic colonization of VAD male mice in all groups combined was significantly higher if the mouse had >10% weight loss than if the mouse had <10% weight loss (S3 Fig). Mock-treated VAD male mice lost significant weight from day 0 to day 4, but none of the



Fig 4. Antibiotic treatment failure occurs at specific concentrations *in vivo*. Enrofloxacin was administered directly into the drinking water on day 2 post-infection with multidrug-resistant S. Typhimurium D23580. Groups of C57BL/6J male (n = 5) and female (n = 5) mice on a standard diet were assessed for the following treatment groups: 0 mg/ml, 0.01 mg/ml, 0.05 mg/ml, and 0.10 mg/ml. Systemic bacterial burden was characterized by CFU in liver, spleen and blood collected at necropsy 5 days after infection (**A**). Water weights and mouse weights were collected daily. The average enrofloxacin consumed ( $\mu$ g/kg/day) was determined for each group based on water and mouse weights for days 2–3, 3–4 and 4–5 (**B**). Data are reported as mean ± SEM. Significance was determined on log-transformed values (**A**) or calculated values (**B**) with a Kruskal-Wallis test and Dunn's multiple comparisons test (\*, p<0.05; ns, not significant).

treatment groups were significant (S4 Fig). In contrast, VAD female mice either showed no significant improvement with any treatment (spleen), or equal effectiveness with enrofloxacin alone and co-treatment (liver, blood) (Fig 5C). In male and female mice on a control diet, both enrofloxacin (0.05 mg/ml) and co-treatment yielded significant or trending decreases in systemic *S*. Typhimurium (S5A and S5B Fig). Since 0.05 mg/ml enrofloxacin was successful, independent of vitamin A, at decreasing *S*. Typhimurium in liver and blood in VAD female mice and control mice for both sexes, a concentration of 0.01 mg/ml was also assessed for these groups. However, co-treatment of 0.01 mg/ml enrofloxacin and vitamin A yielded no benefit in VAD female mice compared to mock-treated animals (S6 Fig). For males on a control diet, co-treatment of 0.01 mg/ml enrofloxacin and vitamin A showed a slight decrease in splenic bacterial levels, but there was no difference in the liver and blood (S7A Fig). There was no benefit of any therapy for females on a control diet (S7B Fig).

Enrofloxacin is a common antibiotic used in veterinary medicine, and it is converted to ciprofloxacin by the liver [26]. No significant differences in plasma enrofloxacin or plasma ciprofloxacin levels were detected between enrofloxacin only and co-treated male (S8A Fig) and



Fig 5. Co-treatment of vitamin A and antibiotics decreases systemic *Salmonella* burden and improves survival in VAD male mice. VAD and control *Slc11a1*<sup>+/+</sup> mice were generated and infected with multidrug resistant *S*. Typhimurium D23580 at 9 weeks of age. Mock-treated mice received 100  $\mu$ l of sterile PBS administered by oral gavage on day 1 and day 2 post-infection. Vitamin A-treated mice received 600 IU of retinyl palmitate in 100  $\mu$ l of sterile PBS administered by oral gavage on day 1 and day 2 post-infection and changed to control diet at day 1 post-infection. The antibiotic enrofloxacin (abx) was administered at day 2 post-infection in the drinking water at 0.05 mg/ml (A). Systemic burden of *S*. Typhimurium was characterized for VAD male (B) and VAD female (C) mice (n = 4–6) as CFU in liver, spleen and blood at necropsy 4 days post-infection. Data are reported as mean ± SEM. Significance was determined on log-transformed values with a Kruskal-Wallis test and Dunn's multiple comparisons test (\*, p<0.05; ns, not significant). (D) Percent survival for control and treated VAD male mice was assessed with an 8-day Kaplan-Meier survival curve (n = 6–16). An enrofloxacin concentration of 0.01 mg/ml was used in drinking water treatment, and oral gavage treatments were given as previously described. Significance was determined with pairwise comparisons between the control group and each other treatment group using a log-rank (Mantel-Cox) test (\*, p<0.05; ns, not significant). Data from untreated controls are also included in Fig 1.

female (<u>S8B Fig</u>) mice, as assessed by LC-MS/MS. These results ruled out the possibility that vitamin A could work by increasing consumption or bioavailability of the antibiotic.

As 0.01 mg/ml was subtherapeutic such that there was still lethal morbidity in VAD male mice, these conditions were used to assess survival of VAD male mice with treatments. Mice were monitored for 8 days post-infection and euthanized upon showing signs of lethal morbidity, and a Kaplan-Meier survival curve was generated (Fig 5D). Of the infected mice fed control diets, 100% of males survived. The percent survival of mock-treated VAD males was only 25%, but this was improved with enrofloxacin only (41.7%) or co-treatment of vitamin A and enrofloxacin (66.7%). Significance between each treatment group and survival of mice on a control diet was assessed with a log-rank (Mantel-Cox) test. Further, for the 44 male mice

represented in the survival curve (Fig 5D), weight loss correlated with survival. Over the course of the experiment, if a mouse lost less than 10% of its weight, it was more likely to have survived (16/21; 76% survival) than if a mouse had lost more than 10% of its weight (7/23; 30% survival). In a separate experiment with a longer time course grouping sexes, vitamin A treatment alone improved survival by 66.7% in VAD mice (S9 Fig). Taken together, these data indicated that vitamin A synergized with subtherapeutic (0.05 mg/ml) enrofloxacin treatment to decrease systemic *S*. Typhimurium levels and improve survival in VAD mice.

#### Discussion

Epidemiologic studies have shown that vitamin A deficiency and disseminated non-typhoidal *Salmonella* infections are co-endemic, disproportionately affecting sub-Saharan Africa [10, 27]. The high mortality rate of disseminated infection has been attributed to antimicrobial resistance of ST313 strains that circulate regionally in sub-Saharan Africa, to genomic features of these strains associated with increased systemic dissemination, and to the high prevalence of co-morbidities that predispose to systemic disease [17, 23, 24]. The current study was designed as a preclinical trial in mice to assess the potential utility of vitamin A as an adjunct to antibiotic therapy. Our results suggest a benefit of combined antibiotic treatment and vitamin A supplementation, with the effect being most apparent in male mice.

It is well-appreciated that vitamin A supplementation can be a useful tool for disease prevention and treatment. In populations with a  $\geq 1\%$  prevalence of night blindness or a  $\geq 20\%$  prevalence of vitamin A deficiency, the WHO recommends vitamin A supplementation for children 12–59 months of age [28]. Vitamin A supplementation was found to significantly reduce the risk of all-cause morbidity and mortality due to diarrhea [29]. In children 2 and under presenting with measles, the WHO-recommended treatment of two consecutive daily megadoses of vitamin A [30] reduces overall and pneumonia-specific mortality [11] and decreases duration of diarrhea and fever. However, the utility of vitamin A for bacterial blood-stream infections has not been assessed. *S.* Typhimurium bloodstream infections typically present as a febrile illness; diarrhea is often absent or not a prominent symptom [10, 31, 32]. Our results suggest that co-treatment with vitamin A and antibiotics may improve outcomes during bloodstream infections with drug-resistant *S.* Typhimurium. Despite WHO's recommendation for vitamin A supplementation, full coverage is still lacking in many countries. Therefore, supplementing with vitamin A during episodes of febrile illness is a potential strategy to increase coverage and improve infection outcomes.

A notable finding of our study was the sex-specific effect of vitamin A supplementation with antibiotics for treatment of *S*. Typhimurium infection in mice. The picture emerging from recent studies of innate immunity is that multiple sex-specific differences impact resistance of males and females to infection. Human males are more susceptible to many infectious diseases, whereas females are more at risk of developing an autoimmune condition [33]. In current events, males are at a higher risk of severe COVID-19 infection [34]. Males also have a higher risk of developing sepsis [35, 36]. A study on sepsis in children found that vitamin A deficiency was more common in the sepsis group as compared to control, and vitamin A deficiency rates were ~80% in the subgroups of severe sepsis and septic shock [37], indicating how a VAD status can predispose to severe disease. The majority of patients with sepsis in this study were male. In addition, some genes encoding for innate immune molecules are located on the X chromosome [33], suggesting a genetic basis for some of the differences observed. It is therefore possible that malnutrition and vitamin A deficiency may compromise the immune response to systemic *Salmonella* infection in a sex-specific manner. We speculate that vitamin A may be playing the role of a host-directed therapy by boosting the immune response of

VAD male mice, synergizing with an initially subtherapeutic dose of antibiotic, and leading to better immunological control of infection. While the current study did not investigate the mechanisms underlying sex differences in the response to vitamin A supplementation, possible explanations include sex differences in innate immune cell function [33], intestinal microbiota composition [38], and hormone levels [39].

Antibiotic treatment failure is the persistence of symptoms despite initiation of antibiotics. Common causes include wrong diagnosis, infection with a resistant organism, or host failure such as malnutrition [25]. We modeled antibiotic treatment failure with a subtherapeutic enrofloxacin concentration that yielded no difference in systemic bacterial burden, but circulating antibiotic was still present in the mouse. It can be the joint action of antibiotics and host defense mechanisms that leads to clearance of bacteria and resolution of disease. While antibiotics are typically studied for their bactericidal or bacteriostatic activity against susceptible bacteria, they can also exist at subtherapeutic or subinhibitory concentrations, i.e. at concentrations below the MIC. Subinhibitory antibiotic concentrations can still affect both the host and the bacterium [40]. It has been proposed that some antibiotics may influence antiphagocytic cell-surface structures and lead to enhanced phagocytosis, modifying immune cell function [41]. For example, neutrophil phagocytosis and intracellular killing activity have been enhanced by pre-incubation with subinhibitory antimicrobial concentrations in some studies, although it depends on the antibiotic used and organism studied [42]. One study reported that subinhibitory levels of fluoroquinolone antibiotics like ciprofloxacin lead to enhanced binding of anti-LPS antibodies to Enterobacteriaceae like Escherichia coli [43]. Taken together, these findings suggest that vitamin A treatment may synergize with subtherapeutic antibiotics to enhance immune function in male mice.

Our study has limitations. We studied the effect of vitamin A co-treatment with only a single antibiotic, enrofloxacin, and a single clinical isolate, D23580. Common treatments for iNTS have included first-line antibiotics of ampicillin, chloramphenicol and trimethoprim/sulfamethoxazole [23]. D23580 is resistant to all of these first-line drugs. It is sensitive to fluoroquinolones at certain concentrations *in vitro* and *in vivo*. However, future studies should incorporate S. Typhimurium strains also resistant to fluoroquinolones, as this is an emerging issue [9]. In addition, it would be important to determine whether vitamin A supplementation can prevent treatment failure with other commonly used antibiotics. There are also limitations to using a mouse model of infection, including the low genetic variation of inbred mice used and the short infection time course permitted by the VAD male mouse model. Future studies could incorporate additional mouse lines, routes of infection and challenge doses. However, given the track record of using vitamin A as prevention and treatment for other infections, this information on whether vitamin A supplementation could be effective against disseminated infections with antibiotic-resistant *Salmonella* might be best obtained in a clinical study.

#### **Supporting information**

S1 Fig. Systemic burden of *Salmonella* in male and female mice on control and VAD diets. A. Data in Fig 1B were re-analyzed to compare *S*. Typhimurium colonization level in each group by sex (n = 7-8). B. Levels of *S*. Typhimurium in liver, spleen and blood at day 5 post-infection in male and female mice (n = 1-7). The same data is represented in two different ways. Initially there were more VAD male mice in the study but they reached lethal endpoint prior to day 5 and were likely to have a higher CFU than in the figure. C. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 and 5 post-infection combined (n = 8-15). Data shown in this figure are a re-analysis of data presented in S1A and S1B. Data represent mean  $\pm$  SEM. Significance between groups was determined with a Mann-Whitney test of logtransformed values (\*, p<0.05; ns, not significant). (TIF)

S2 Fig. Systemic burden of *Salmonella* in male and female mice during antibiotic treatment. Relating to Fig 4A, comparison of levels of *S*. Typhimurium in liver, spleen and blood at day at day 5 post-infection in male and female mice on a standard diet after antibiotic treatment (n = 5). Data represent mean  $\pm$  SEM. Significance between male and female mice was determined with a Mann-Whitney test of log-transformed values (\*, p<0.05; ns, not significant).

(TIF)

S3 Fig. Systemic burden of *Salmonella* as compared to percent weight lost by VAD male mice during infection. Relating to VAD male mice in Fig 5B, systemic colonization of mice in all treatment groups combined grouped by <10% weight loss (n = 14) versus >10% weight loss (n = 12) by day 4 post-infection with *S*. Typhimurium D23580. Data represent mean  $\pm$  SEM. Significance between weight loss groups was determined with a Mann-Whitney test of log-transformed values. A p<0.05 was considered significant. (TIF)

S4 Fig. Weight change of mock-treated and treated VAD male mice during infection with *Salmonella*. Relating to VAD male mice (n = 4–6) in Fig 5B, weight change during infection with *S*. Typhimurium D23580 during mock treatment, treatment with vitamin A only, treatment with 0.05 mg/ml enrofloxacin in the drinking water only, or vitamin A and antibiotic co-treatment. Data represent mean  $\pm$  SEM. Significance between day 0 and day 4 weights was determined with a Mann-Whitney test (\*, p<0.05; ns, not significant). (TIF)

S5 Fig. Systemic burden of *Salmonella* in male and female mice on a control diet during treatment with 0.05 mg/ml enrofloxacin. *A*. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in male mice (n = 3-6) on a control diet administered either mock treatment, vitamin A only, 0.05 mg/ml enrofloxacin only, or co-treatment. *B*. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in female mice (n = 4-7) on a control diet administered either mock treatment, vitamin A only, 0.05 mg/ml enrofloxacin only, 0.05 mg/ml enrofloxacin only, or co-treatment. *B*. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in female mice (n = 4-7) on a control diet administered either mock treatment, vitamin A only, 0.05 mg/ml enrofloxacin only, or co-treatment. Data represent mean ± SEM. Significance was determined on log-transformed values with a Kruskal-Wallis test and Dunn's multiple comparisons test (\*, p < 0.05; ns, not significant). Data from untreated control mice are also shown in Fig 1. (TIF)

S6 Fig. Systemic burden of *Salmonella* in female mice on a VAD diet during treatment with 0.01 mg/ml enrofloxacin. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in female mice (n = 4–6) on a VAD diet administered either mock treatment, vitamin A only, 0.01 mg/ml enrofloxacin only, or co-treatment. VAD males treated with 0.01 mg/ml showed decreased survival and are not shown. Data represent mean  $\pm$  SEM. Significance was determined on log-transformed values with a Kruskal-Wallis test and Dunn's multiple comparisons test (\*, p<0.05; ns, not significant). (TIF)

S7 Fig. Systemic burden of *Salmonella* in male and female mice on a control diet during treatment with 0.01 mg/ml enrofloxacin. A. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in male mice (n = 3-4) on a control diet administered either mock treatment, vitamin A only, 0.01 mg/ml enrofloxacin only, or co-treatment. B. Levels of *S*. Typhimurium in liver, spleen and blood at day 4 post-infection in female mice (n = 4-7) on

a control diet administered either mock treatment, vitamin A only, 0.01 mg/ml enrofloxacin only, or co-treatment. Data represent mean  $\pm$  SEM. Significance was determined on log-transformed values with a Kruskal-Wallis test and Dunn's multiple comparisons test (\*, p<0.05; ns, not significant).

(TIF)

S8 Fig. Concentrations of antibiotic in the plasma during treatment with 0.05 mg/ml enrofloxacin. *A*. Levels of enrofloxacin and ciprofloxacin (ng/ml) in the plasma during mock treatment, 0.05 mg/ml enrofloxacin in the drinking water, or co-treatment of 0.05 mg/ml enrofloxacin and vitamin A in male mice (n = 1–2) on control or VAD diet, as assessed by LC-MS/MS. *B*. Levels of enrofloxacin and ciprofloxacin (ng/ml) in the plasma during mock treatment, 0.05 mg/ml enrofloxacin in the drinking water, or co-treatment of 0.05 mg/ml enrofloxacin and vitamin A in female mice (n = 3–4) on control or VAD diet, as assessed by LC-MS/MS. Data represent mean ± SEM. Significance between antibiotic levels in antibiotictreated and co-treated mice was determined with a Mann-Whitney test (\*, p<0.05; ns, not significant).

(TIF)

**S9 Fig. Survival of mice when treated with vitamin A only over a 13-day time course.** Survival of grouped male and female mice treated with two consecutive doses of either PBS or retinyl palmitate (600 IU delivered by oral gavage) starting 1 day after *S*. Typhimurium infection. Additionally, VAD mice treated with retinyl palmitate were placed on a control diet replete with vitamin A one-day after *S*. Typhimurium infection. Data represent percent survival of 12 mice per group from three independent experiments. Pairwise comparisons between the control diet group and each other group was determined with a survival analysis log-rank (Mantel-Cox) test (\*, p<0.05). (TIF)

Acknowledgments

The authors thank Greg Barton for providing inbred C57BL/6-*Slc11a1*<sup>+/+</sup> mice, Robert Heyderman and the Malawi-Liverpool-Wellcome Trust Clinical Research Centre for providing the *S*. Typhimurium clinical isolate D23580, Charles Stephensen for advice on establishing the vitamin A-deficient mouse model, and Michael Rogawski for input on enrofloxacin measurements by LC-MS/MS.

#### **Author Contributions**

Conceptualization: Annica R. Stull-Lane, Kristen L. Lokken-Toyli.

Formal analysis: Annica R. Stull-Lane, Kristen L. Lokken-Toyli.

Funding acquisition: Renée M. Tsolis.

**Investigation:** Annica R. Stull-Lane, Vladimir E. Diaz-Ochoa, Gregory T. Walker, Stephanie A. Cevallos, Andromeda L. N. Winter, Ariel Del Hoyo Muñoz, Guiyan G. Yang, Eric M. Velazquez, Chun-Yi Wu.

Methodology: Annica R. Stull-Lane, Kristen L. Lokken-Toyli, Chun-Yi Wu.

Supervision: Renée M. Tsolis.

Writing - original draft: Annica R. Stull-Lane.

Writing – review & editing: Annica R. Stull-Lane, Kristen L. Lokken-Toyli, Gregory T. Walker, Stephanie A. Cevallos, Andromeda L. N. Winter, Ariel Del Hoyo Muñoz, Guiyan G. Yang, Eric M. Velazquez, Chun-Yi Wu, Renée M. Tsolis.

#### References

- 1. Stephensen CB. Burden of infection on growth failure. Journal of Nutrition. 1999; 129(2):534s-8s.
- Muller O, Krawinkel M. Malnutrition and health in developing countries. Canadian Medical Association Journal. 2005; 173(3):279–86. https://doi.org/10.1503/cmaj.050342 PMID: 16076825
- Mitra AK, Alvarez JO, Stephensen CB. Increased urinary retinol loss in children with severe infections. Lancet. 1998; 351(9108):1033–4. https://doi.org/10.1016/S0140-6736(05)79000-0 PMID: 9546517
- Stephensen CB. Vitamin A, infection, and immune function. Annual Review of Nutrition. 2001; 21:167– 92. https://doi.org/10.1146/annurev.nutr.21.1.167 PMID: 11375434
- Humphries RM, Schuetz AN. Antimicrobial susceptibility testing of bacteria that cause gastroenteritis. Clinics in Laboratory Medicine. 2015; 35(2):313–31. <u>https://doi.org/10.1016/j.cll.2015.02.005</u> PMID: 26004645
- Majowicz SE, Musto J, Scallan E, Angulo FJ, Kirk M, O'Brien SJ, et al. The global burden of nontyphoidal Salmonella gastroenteritis. Clinical Infectious Diseases. 2010; 50(6):882–9. <u>https://doi.org/10.1086/</u> 650733 PMID: 20158401
- Ao TT, Feasey NA, Gordon MA, Keddy KH, Angulo FJ, Crump JA. Global burden of invasive nontyphoidal Salmonella disease, 2010(1). Emerg Infect Dis. 2015; 21(6). Epub 2015/04/11. <u>https://doi.org/10.3201/eid2106.140999</u> PMID: 25860298.
- Balasubramanian R, Im J, Lee JS, Jeon HJ, Mogeni OD, Kim JH, et al. The global burden and epidemiology of invasive non-typhoidal Salmonella infections. Hum Vaccin Immunother. 2018:1–6. Epub 2018/ 08/08.
- Chattaway MA, Aboderin AO, Fashae K, Okoro CK, Opintan JA, Okeke IN. Fluoroquinolone-resistant enteric bacteria in sub-Saharan Africa: clones, implications and research needs. Frontiers in Microbiology. 2016; 7. ARTN 558. https://doi.org/10.3389/fmicb.2016.00558 PMID: 27148238
- Feasey NA, Dougan G, Kingsley RA, Heyderman RS, Gordon MA. Invasive non-typhoidal Salmonella disease: an emerging and neglected tropical disease in Africa. Lancet. 2012; 379(9835):2489–99. https://doi.org/10.1016/S0140-6736(11)61752-2 PMID: 22587967.
- Huiming YM, Chaomin WM, Meng M. Vitamin A for treating measles in children. Cochrane Db Syst Rev. 2005;(4). ARTN CD001479.pub3. <u>https://doi.org/10.1002/14651858.CD001479.pub3</u> PMID: 16235283
- 12. Guideline: Updates on the Management of Severe Acute Malnutrition in Infants and Children. WHO Guidelines Approved by the Guidelines Review Committee. Geneva: World Health Organization. 2013.
- 13. Murray JS, White J. Vitamin A supplementation in infants and children. J Spec Pediatr Nurs. 2016; 21 (4):212–7. https://doi.org/10.1111/jspn.12156 PMID: 27596119
- Twining SS, Schulte DP, Wilson PM, Fish BL, Moulder JE. Vitamin A deficiency alters rat neutrophil function. Journal of Nutrition. 1997; 127(4):558–65. <u>https://doi.org/10.1093/jn/127.4.558</u> PMID: 9109605
- Kaufmann SHE, Dorhoi A, Hotchkiss RS, Bartenschlager R. Host-directed therapies for bacterial and viral infections. Nat Rev Drug Discov. 2018; 17(1):35–56. https://doi.org/10.1038/nrd.2017.162 PMID: 28935918
- Arpaia N, Godec J, Lau L, Sivick KE, McLaughlin LM, Jones MB, et al. TLR signaling is required for Salmonella Typhimurium virulence. Cell. 2011; 144(5):675–88. <u>https://doi.org/10.1016/j.cell.2011.01.031</u> PMID: 21376231
- Singletary LA, Karlinsey JE, Libby SJ, Mooney JP, Lokken KL, Tsolis RM, et al. Loss of multicellular behavior in epidemic African nontyphoidal Salmonella enterica serovar Typhimurium ST313 strain D23580. Mbio. 2016; 7(2). ARTN e02265-15. https://doi.org/10.1128/mBio.02265-15 PMID: 26933058
- 18. M100: Performance Standards for Antimicrobial Susceptibility Testing. 27th Edition ed. Clinical and Laboratory Standards Institute. 2017.
- Rankin JD, Taylor RJ. The estimation of doses of Salmonella typhimurium suitable for the experimental production of disease in calves. Vet Rec. 1966; 78(21):706–7. Epub 1966/05/21. <u>https://doi.org/10. 1136/vr.78.21.706 PMID: 5336163.</u>
- Beltran P, Plock SA, Smith NH, Whittam TS, Old DC, Selander RK. Reference collection of strains of the Salmonella typhimurium complex from natural populations. J Gen Microbiol. 1991; 137(3):601–6. Epub 1991/03/01. https://doi.org/10.1099/00221287-137-3-601 PMID: 2033380.

- Fu S, Octavia S, Tanaka MM, Sintchenko V, Lan R. Defining the Core Genome of Salmonella enterica Serovar Typhimurium for Genomic Surveillance and Epidemiological Typing. J Clin Microbiol. 2015; 53 (8):2530–8. Epub 2015/05/29. https://doi.org/10.1128/JCM.03407-14 PMID: 26019201.
- Hoffmann R, Schneider A, Quamo Y. The sex difference in vitamin-A Metabolism. Journal of Investigative Dermatology. 1950; 15(6):409–19. https://doi.org/10.1038/jid.1950.123 PMID: 14794994
- Cuypers WL, Jacobs J, Wong V, Klemm EJ, Deborggraeve S, Van Puyvelde S. Fluoroquinolone resistance in Salmonella: insights by whole-genome sequencing. Microb Genomics. 2018; 4(7). ARTN 000195. https://doi.org/10.1099/mgen.0.000195 PMID: 29975627
- Kingsley RA, Msefula CL, Thomson NR, Kariuki S, Holt KE, Gordon MA, et al. Epidemic multiple drug resistant Salmonella Typhimurium causing invasive disease in sub-Saharan Africa have a distinct genotype. Genome Research. 2009; 19(12):2279–87. https://doi.org/10.1101/gr.091017.109 PMID: 19901036
- Grant GB, Campbell H, Dowell SF, Graham SM, Klugman KP, Mulholland EK, et al. Recommendations for treatment of childhood non-severe pneumonia. Lancet Infect Dis. 2009; 9(3):185–96. https://doi.org/ 10.1016/S1473-3099(09)70044-1 PMID: 19246022
- Martinez M, McDermott P, Walker R. Pharmacology of the fluoroquinolones: a perspective for the use in domestic animals. Vet J. 2006; 172(1):10–28. <u>https://doi.org/10.1016/j.tvjl.2005.07.010</u> PMID: 16154368
- Global prevalence of vitamin A deficiency in populations at risk 1995–2005: WHO global database on vitamin A deficiency. Geneva: World Health Organization. 2009.
- 28. Guideline: Vitamin A Supplementation in Infants and Children 6–59 Months of Age. WHO Guidelines Approved by the Guidelines Review Committee. Geneva: World Health Organization. 2011.
- Imdad A, Mayo-Wilson E, Herzer K, Bhutta ZA. Vitamin A supplementation for preventing morbidity and mortality in children from six months to five years of age. Cochrane Database Syst Rev. 2017; 3: CD008524. Epub 2017/03/11. https://doi.org/10.1002/14651858.CD008524.pub3 PMID: 28282701.
- **30.** Treating measles in children. Department of Immunization, Vaccines and Biologicals. Geneva: World Health Organization. 2004.
- Gordon MA, Banda HT, Gondwe M, Gordon SB, Boeree MJ, Walsh AL, et al. Non-typhoidal Salmonella bacteraemia among HIV-infected Malawian adults: high mortality and frequent recrudescence. Aids. 2002; 16(12):1633–41. https://doi.org/10.1097/00002030-200208160-00009 PMID: 12172085
- Graham SM, Walsh AL, Molyneux EM, Phiri AJ, Molyneux ME. Clinical presentation of non-typhoidal Salmonella bacteraemia in Malawian children. T Roy Soc Trop Med H. 2000; 94(3):310–4. https://doi. org/10.1016/S0035-9203(00)90337-7
- Jaillon S, Berthenet K, Garlanda C. Sexual dimorphism in innate immunity. Clin Rev Allerg Immu. 2019; 56(3):308–21. https://doi.org/10.1007/s12016-017-8648-x PMID: 28963611
- Sharma G, Volgman AS, Michos ED. Sex Differences in Mortality from COVID-19 Pandemic: Are Men Vulnerable and Women Protected? JACC Case Rep. 2020. Epub 2020/05/07. <u>https://doi.org/10.1016/j.jaccas.2020.04.027 PMID: 32373791.</u>
- Kisat M, Villegas CV, Onguti S, Zafar SN, Latif A, Efron DT, et al. Predictors of sepsis in moderately severely injured patients: an analysis of the National Trauma Data Bank. Surg Infect. 2013; 14(1):62–8. https://doi.org/10.1089/sur.2012.009 PMID: 23461696
- Schroder J, Kahlke V, Staubach KH, Zabel P, Stuber F. Gender differences in human sepsis. Arch Surg-Chicago. 1998; 133(11):1200–4. <u>https://doi.org/10.1001/archsurg.133.11.1200</u> PMID: 9820351
- Zhang X, Yang K, Chen L, Liao X, Deng L, Chen S, et al. Vitamin A deficiency in critically ill children with sepsis. Crit Care. 2019; 23(1):267. Epub 2019/08/03. <u>https://doi.org/10.1186/s13054-019-2548-9</u> PMID: 31370866.
- Huda MN, Ahmad SM, Kalanetra KM, Taft DH, Alam MJ, Khanam A, et al. Neonatal Vitamin A Supplementation and Vitamin A Status Are Associated with Gut Microbiome Composition in Bangladeshi Infants in Early Infancy and at 2 Years of Age. J Nutr. 2019; 149(6):1075–88. Epub 2019/04/23. <a href="https://doi.org/10.1093/jn/nxz034">https://doi.org/10.1093/jn/nxz034</a> PMID: 31006815.
- Hines M, Spencer D, Kung KT, Browne WV, Constantinescu M, Noorderhaven RM. The early postnatal period, mini-puberty, provides a window on the role of testosterone in human neurobehavioural development. Curr Opin Neurobiol. 2016; 38:69–73. Epub 2016/03/15. https://doi.org/10.1016/j.conb.2016. 02.008 PMID: 26972372.
- Li J, Xie SY, Ahmed S, Wang FN, Gu YF, Zhang CN, et al. Antimicrobial Activity and Resistance: Influencing Factors. Front Pharmacol. 2017; 8. ARTN 364. <u>https://doi.org/10.3389/fphar.2017.00364</u> PMID: 28659799

- **41.** Overbeek BP, Veringa EM. Role of Antibodies and Antibiotics in Aerobic Gram-Negative Septicemia— Possible Synergism between Antimicrobial Treatment and Immunotherapy. Reviews of Infectious Diseases. 1991; 13(4):751–60. https://doi.org/10.1093/clinids/13.4.751 PMID: 1925294
- Zhanel GG, Hoban DJ, Harding GK. Subinhibitory antimicrobial concentrations: A review of in vitro and in vivo data. Can J Infect Dis. 1992; 3(4):193–201. Epub 1992/07/01. https://doi.org/10.1155/1992/ 793607 PMID: 22514370.
- Nelson D, Delahooke TES, Poxton IR. Influence of Subinhibitory Levels of Antibiotics on Expression of Escherichia-Coli Lipopolysaccharide and Binding of Antilipopolysaccharide Monoclonal-Antibodies. Journal of Medical Microbiology. 1993; 39(2):100–6. <u>https://doi.org/10.1099/00222615-39-2-100</u> PMID: 8345504